Author’s response to reviews

Title: SLC5A2 mutations, including two novel mutations, responsible for renal glucosuria in Chinese families

Authors:

Lei Yu (lanlife@163.com)
MENG WU (mengwu2018ut@163.com)
PING HOU (houp505@126.com)
HONG ZHANG (hongzh@bjmu.edu.cn)

Version: 2 Date: 20 Aug 2019

Author’s response to reviews:

Dear editor,

Thank you for your letter and for the editorial comments concerning our manuscript entitled "SLC5A2 mutations including two novel mutations responsible for Chinese renal glucosuria families: A case report" (BNEP-D-19-00559R1). We do appreciate greatly for the comments and suggestions by the Editorial Board. We have revised the manuscript and responded to the comments point-by-point (See the following responses to editor). The manuscript has been read and approved by all of the authors.

Sincerely yours,

Lei Yu,

Renal Division
Inner Mongolia People’s Hospital
Hohhot, 010017
P.R. China
Response to editor

1. Formatting

A1: We do appreciate the valuable comments and nice suggestions. We reformatted our manuscript into a research article and changed the article type.

2. Ethics approval and consent to participate

A2: Thank you for the suggestions. We declared in the manuscript: The study protocol was approved by the Medical Ethics Committee of Inner Mongolia People's Hospital (Hohhot, China). Written informed consent for taking part in the study was obtained from all participants prior to participation in the study.

3. Consent for publication

Thank you for the suggestion. We declared in the manuscript: Written informed consent for publication was obtained from the patients.